\documentclass[11pt,a4paper]{ivoa}
\input tthdefs

\usepackage[utf8]{inputenc}
\usepackage{booktabs, tabulary}    % for nicer tables
\usepackage[titletoc,toc,title]{appendix}

% make the text in pdf properly searchable
\usepackage{lmodern}

% use listings for including text files and code snippets
\usepackage{listings}
\usepackage{color}
\usepackage[nottoc]{tocbibind}

\definecolor{mygreen}{rgb}{0,0.6,0}
\definecolor{mygray}{rgb}{0.5,0.5,0.5}
\definecolor{mymauve}{rgb}{0.58,0,0.82}

\ivoagroup{DM}
\author{François Bonnarel}
\author{Mireille Louys}
\author{Markus Nullmeier}
\author{Kristin Riebe}
\author{Michèle Sanguillon}
\author{Mathieu Servillat}
\author{IVOA Data Model Working Group}

\editor{François Bonnarel}

% smaller font in verbatim:
\usepackage{verbatimbox}

% for nicer tables:
\renewcommand{\arraystretch}{1.3}
\newcommand{\head}[1]{\textbf{#1}}

% define new command for classes and other stuff, in case we decide later on for a different style
\newcommand{\class}[1]{\emph{#1}}
\newcommand{\attribute}[1]{\texttt{#1}}
\newcommand{\attr}[1]{\texttt{#1}} % equivalent to \attribute

\newcommand{\urlparam}[1]{{\scriptsize{#1}}}
\newcommand{\urlvalue}[1]{{\scriptsize{#1}}}

% from https://tex.stackexchange.com/questions/121955/help-on-dealing-with-items-divided-with-slash
\renewcommand{\slash}{/\penalty\exhyphenpenalty\hspace{0pt}}

\begin{document}
\newcolumntype{Y}{>{\raggedright\arraybackslash}X}

\title{IVOA Provenance Table Access Protocol (ProvTAP)}

\begin{abstract}
This document describes the ProvTAP protocol for accessing provenance information according to the IVOA ProvenanceDM standard. It defines how the elements of ProvDM are described in the TAP schema tables and provides guidelines for implemnienting with TAP 1.1.
\end{abstract}

\section{Introduction}
\subsection{context}
ProvTAP is a TAP service specialization for delivering IVOA Provenance metadata an build up serialisation documents consistently with the IVOA Provenance data model \citep{std:ProvenanceDM}
As for any TAP service \citep{std:TAP}, ProvTAP is providing ADQL query responses
consisting of a single table each, gathering columns selected from various
tables defined in the service's TAP schema. The  default format of the Query Response is VOTable. Other formats (JSON, tsv, etc.) lay also be optionallyi supported.

The definition of a ProvTAP service is similar to the approach used in the ObsCore specification \citep{std:ObsCore} where ObsCore data model plays the role of underlying model for ObsTAP services. It is also inspired by the multiple table 
design proposed in the the RegTAP specification \citep{std:RegTAP} with respect to VOResource data model.

Basically, data-model consistency for a TAP service is asserted by the datamodel element in the capability of the service.  The relational view of the model is expressed via the table structure in the TAP schema and the definition of each column datatype, ucd, utype attributes  in the same schema.

ProvTAP is not the only DAL service able to provide provenance metadata. The ProvDAL \citep{spec:ProvDAL}  specification offers a complementary method based on a search-by-parameters strategy and does not require the server to adopt the TAP architecture. ProvDAL services allow to retrievxes sequences or subgraphs of Provenance metadata associated to a starting entity or activity allowing different combinations. 

\subsection{Role within the VO architecture}

IVOA ProvTAP specification is used to define TAP services with a specific TAP schema expressing view of the IVOA provenance Data Model. It relies on DALI for general DAL definitions.  It
benefits from existing IVOA notations and standards like UCD, VOUnits, VOTable, VOResource  and extensions of the latter such as VODataservice and  TAPREgExt.

\begin{figure}
\centering
\includegraphics[width=0.9\textwidth]{VOArchitecture-ProvTAP2018.png}
\caption[Architecture diagram for the ProvTAP specification]{Architecture diagram for the ProvTAP specification. It is based on a view of  IVOA Provenance data model an relies on TAP access technology. As such it is fully integrated in the IVOA framework}
\label{fig:archdiag}
\end{figure}

Fig.~\ref{fig:archdiag} shows the dependencies of this document with respect to other existing standards.


\section{ProvTAP in action}




Here are a few examples of ADQL queries for ProvTAP (values are taken from the test metadata presented in appendix) :
\begin{itemize}
\item To retrieve all activity metadata for activies sharing the same activityDescription, by looking for the activities with the same a\_description value: \\
\texttt{SELECT * FROM  Activity WHERE \\
  Activity.a\_description = 'hipsgend\_mean'}
\item To retrieve all activities associated with a specific agent: \\
\texttt{SELECT WasAssociatedWith.waw\_activity\_id, Activity.a\_name, \\}
\verb!       !\texttt{Activity.a\_annotation FROM WasAssociatedWith \\
INNER JOIN Activity ON \\ WasAssociatedWith.waw\_activity\_id = Activity.a\_id
\\ WHERE  WasAssociatedWith.waw\_agent\_id = 'obspm'}
\item To retrieve all entities identifiers attributed to \texttt{curator} agents: \\
\texttt{SELECT WasAttributedTo.wat\_entity\_id FROM WasAttributedTo \\
 WHERE WasAttributedTo.wat\_role = 'curator'}
\end{itemize}
%\TODO{Do we need combined query possibilities, i.e. ask for ObsCore-fields and Provenance fields in one query? Or rather use a 2-step-process, decoupling them from each other?}

%\TODO{Also look at PROV-AQ from the W3C.}

The result of the first ADQL query is the following.
\begin{verbnobox}[\scriptsize]
<?xml version="1.0" encoding="UTF-8"?>
<VOTABLE version="1.2" xmlns="http://www.ivoa.net/xml/VOTable/v1.2" 
  xmlns:ex="http://www.example.com/provenance" 
  xmlns:ivo="http://www.ivoa.net/documents/rer/ivo/"
  xmlns:voprov="http://www.ivoa.net/documents/dm/provdm/voprov/" 
  xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" 
  xsi:schemaLocation="http://www.ivoa.net/xml/VOTable/v1.2 http://www.ivoa.net/xml/VOTable/VOTable-1.2.xsd">
  <RESOURCE type="provenance">
    <DESCRIPTION>Provenance VOTable</DESCRIPTION>
    <TABLE >
      <FIELD id="a_ID" name="a_id" ucd="meta.id" utype="voprov:Activity.id" arraysize="*" datatype="char"/>
      <FIELD id="a_nameID" name="a_name" ucd="meta.title" utype="voprov:Activity.name" arraysize="*" datatype="char"/>
      <FIELD id="a_startID" name="a_startTime" ucd="time.start" utype="voprov:Activity.startTime" arraysize="*" datatype="char" />
      <FIELD id="a_endID" name="a_endTime" ucd="time.start" utype="voprov:Activity.endTime" arraysize="*" datatype="char"/>
      <FIELD id="a_annotID" name="a_annotation" ucd="meta.description" utype="voprov:Activity.annotation" arraysize="*" datatype="char"/>
      <FIELD id="a_descID"  name="a_description" ucd="meta.id" utype="voprov:Activity.description" datatype="char" arraysize="*"/>
      <DATA>
        <TABLEDATA>
          <TR>
            <TD>HiPSgen1</TD>
            <TD>HiPSgen 1</TD>
            <TD>2017-04-18 17:28:00</TD>
            <TD>2017-04-19 17:29:00</TD>
            <TD>HiPS generation launched on the 18th of April 2017 interpolation method </TD>
            <TD>hipsgend_mean</TD>
          </TR>
        </TABLEDATA>
      </DATA>
    </TABLE>

    <INFO name="QUERY_STATUS" value="OK"/>
  </RESOURCE>
</VOTABLE>

\end{verbnobox}

\section{VOSI availability and capabilities}
According to the DALI specification for VO services \citep{std:DALI}, a provenance service implementing ProvTAP must provide a VOSI availability interface as well as a capabilities interface with entries for ProvTAP. The \texttt{standardId}s for these provenance interfaces are:

\begin{verbatim}
ivo://ivoa.net/std/ProvenanceDM#ProvTAP-1.0
\end{verbatim}

The capability for a TAP service to support the Provenance DM is expressed by
the dataModel element as:
\begin{verbatim}
<dataModel ivoid="ivo://ivoa.net/std/ProvenanceDM-1.0">
  ProvenanceDM-1.0
</dataModel>
\end{verbatim}

For ProvTAP, the VOSI tables interface must also be provided.


\section{VOTable serialization for provenance metadata and ProvTAP TAP schema}
The table and column definitions available in the TAP schema, including their
metadata (\attribute{unit}, \attribute{ucd}, \attribute{utype}, \attribute{datatype}, \attribute{description}) exactly  define
 the VOTable mapping of the IVOA Provenance data model (see table and
column definitions and metadata.
\label{sec:appendix-prov-votable}
\input{prov-votable}

\begin{appendices}

\section{TAP schema in xml}
\begin{lstlisting}[language=XML, caption= TAP schema in xml]

<?xml version="1.0" encoding="UTF-8"?>
<vosi:tableset xmlns:vosi="http://www.ivoa.net/xml/VOSITables/v1.0" xmlns:vod="http://www.ivoa.net/xml/VODataService/v1.1" xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:schemaLocation="http://www.ivoa.net/xml/VODataService/v1.1 http://www.ivoa.net/xml/VODataService/v1.1 http://www.ivoa.net/xml/VOSITables/v1.0 http://vo.ari.uni-heidelberg.de/docs/schemata/VOSITables-v1.0.xsd">
  <schema>
    <name>provenance</name>
    <description>Provenance schema</description>
    <table type="output">
      <name>Entity</name>
      <description>instances of Entity class</description>
      <column>
        <name>e_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Entity.id</utype>  
      </column>
      <column>
        <name>e_name</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.title</ucd>
        <utype>voprov:Entity.name</utype> 
      </column>
      <column>
        <name>e_type</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:Entity.type</utype> 
      </column>
      <column>
        <name>e_rights</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:Entity.rights</utype>  
      </column>
      <column>
        <name>e_annotation</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.description</ucd>
        <utype>voprov:Entity.annotation</utype>
      </column>
      <column>
        <name>e_hadMember</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.member</ucd>
        <utype>voprov:Entity.hadMember</utype>
      </column>
      <column>
        <name>e_description</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Entity.description</utype>
      </column>
      <foreignKey>
        <targetTable>EntityDescription</targetTable>
        <fkColumn>
          <fromColumn>e_description</fromColumn>
          <targetColumn>ed_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>EntityDescription</name>
      <description>instances of Entity Description class</description>
      <column>
        <name>ed_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:EntityDescription.id</utype>  
      </column>
      <column>
        <name>ed_name</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.title</ucd>
        <utype>voprov:EntityDescription.name</utype> 
      </column>
      <column>
        <name>ed_doculink</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.ref.url</ucd>
        <utype>voprov:EntityDescription.doculink</utype>  
      </column>
      <column>
        <name>ed_annotation</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.description</ucd>
        <utype>voprov:EntityDescription.annotation</utype>
      </column>
      <column>
        <name>ed_category</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:EntityDescription.category</utype>
      </column>
    </table>
    <table type="output">
      <name>Activity</name>
      <description>instances of Activity  class</description>
      <column>
        <name>a_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Activity.id</utype>  
      </column>
      <column>
        <name>a_name</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.title</ucd>
        <utype>voprov:Activity.name</utype> 
      </column>

      <column>
        <name>a_startTime</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>time.start</ucd>
        <utype>voprov:Activity.startTime</utype>  
      </column>
      <column>
        <name>a_endTime</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>time.end</ucd>
        <utype>voprov:Activity.endTime</utype>  
      </column>
      <column>
        <name>a_annotation</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.description</ucd>
        <utype>voprov:Activity.annotation</utype>
      </column>
      <column>
        <name>a_votype</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:Activity.votype</utype> 
      </column>
      <column>
        <name>a_hadStep</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.member</ucd>
        <utype>voprov:Activity.hadStep</utype>
      </column>
      <column>
        <name>a_description</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Activity.description</utype>
      </column>
      <foreignKey>
        <targetTable>ActivityDescription</targetTable>
        <fkColumn>
          <fromColumn>a_description</fromColumn>
          <targetColumn>ad_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>ActivityDescription</name>
      <description>instanceof Activity Descriptions </description>
      <column>
        <name>ad_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:ActivityDescription.id</utype>  
      </column>
      <column>
        <name>ad_name</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.title</ucd>
        <utype>voprov:ActivityDescription.name</utype> 
      </column>
      <column>
        <name>ad_doculink</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.ref.url</ucd>
        <utype>voprov:ActivityDescription.doculink</utype>  
      </column>
      <column>
        <name>ad_annotation</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.description</ucd>
        <utype>voprov:ActivityDescription.annotation</utype>
      </column>
      <column>
        <name>ad_type</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:ActivityDescription.type</utype>
      </column>
      <column>
        <name>ad_subtype</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:ActivityDescription.subtype</utype>
      </column>
    </table>
    <table type="output">
      <name>Agent</name>
      <description>instance of Agent class </description>
      <column>
        <name>ag_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Agent.id</utype>  
      </column>
      <column>
        <name>ag_name</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.title</ucd>
        <utype>voprov:Agent.name</utype> 
      </column>
      <column>
        <name>ag_type</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:Agent.type</utype>
      </column>
      <column>
        <name>ag_address</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.address</ucd>
        <utype>voprov:Agent.address</utype>
      </column>
      <column>
        <name>ad_email</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.email</ucd>
        <utype>voprov:Agent.email</utype>
      </column>
    </table>
    <table type="output">
      <name>Parameter</name>
      <description>instance of Parameter class </description>
      <column>
        <name>p_isaParamOf</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Parameter.Activity</utype>  
      </column>
      <column>
        <name>p_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Parameter.id</utype>  
      </column>
      <column>
        <name>p_value</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>stat.value</ucd>
        <utype>voprov:Parameter.value</utype> 
      </column>
      <column>
        <name>p_description</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Parameter.description</utype>
      </column>
      <foreignKey>
        <targetTable>ParameterDescription</targetTable>
        <fkColumn>
          <fromColumn>p_description</fromColumn>
          <targetColumn>pd_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
        <targetTable>Activity</targetTable>
        <fkColumn>
          <fromColumn>p_isaParamOf/fromColumn>
          <targetColumn>a_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>ParameterDescription</name>
      <description>instance of Parameter Descriptions </description>
      <column>
        <name>pd_isaParamOf</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:ParameterDescription.ActivityDescription</utype>  
      </column>
      <column>
        <name>pd_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:ParameterDescription.id</utype>  
      </column>
      <column>
        <name>pd_name</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.title</ucd>
        <utype>voprov:ParameterDescription.name</utype> 
      </column>
      <column>
        <name>pd_annotation</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.description</ucd>
        <utype>voprov:ParameterDescription.annotation</utype>
      </column>
      <column>
        <name>pd_datatype</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta</ucd>
        <utype>voprov:ParameterDescription.datatype</utype>
      </column>
      <column>
        <name>pd_unit</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.unit</ucd>
        <utype>voprov:ParameterDescription.unit</utype>
      </column>
      <column>
        <name>pd_ucd</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.ucd</ucd>
        <utype>voprov:ParameterDescription.ucd</utype>
      </column>
      <column>
        <name>pd_utype</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta</ucd>
        <utype>voprov:ParameterDescription.utype</utype>
      </column>
      <column>
        <name>pd_utype</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta</ucd>
        <utype>voprov:ParameterDescription.utype</utype>
      </column>
      <column>
        <name>pd_min</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>stat.min</ucd>
        <utype>voprov:ParameterDescription.min</utype>
      </column>
      <column>
        <name>pd_max</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>stat.max</ucd>
        <utype>voprov:ParameterDescription.max</utype>
      </column>
      <column>
        <name>pd_options</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta</ucd>
        <utype>voprov:ParameterDescription.options</utype>
      </column>  
      <foreignKey>
        <targetTable>ActivityDescription</targetTable>
        <fkColumn>
          <fromColumn>pd_isaParamOf/fromColumn>
          <targetColumn>ad_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>Used</name>
      <description>instance of Used class</description>
      <column>
        <name>u_entity_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Used.entity</utype>  
      </column>
      <column>
        <name>u_activity_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Used.activity</utype> 
      </column>

      <column>
        <name>u_usedDescription_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:Used.usedDescription</utype>
      </column>
      <column>
        <name>u_time</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>time.start</ucd>
        <utype>voprov:Used.time</utype>
      </column>
      <foreignKey>
        <targetTable>UsedDescription</targetTable>
        <fkColumn>
          <fromColumn>u_usedDescription</fromColumn>
          <targetColumn>ud_usedDescription_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
        <targetTable>Entity</targetTable>
        <fkColumn>
          <fromColumn>u_entity_id</fromColumn>
          <targetColumn>e_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
        <targetTable>Activity</targetTable>
        <fkColumn>
          <fromColumn>u_activity_id</fromColumn>
          <targetColumn>a_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>UsedDescription</name>
      <description>instance of Used Descriptions</description>
      <column>
        <name>ud_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:UsedDescription.id</utype>  
      </column>
      <column>
        <name>ud_entityDescription_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:UsedDescription.entityDescription</utype>  
      </column>
      <column>
        <name>ud_activityDescription_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:UsedDescription.activityDescription</utype> 
      </column>
      <column>
        <name>ud_role</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:UsedDescription.entityRole</utype>
      </column>
      <foreignKey>
        <targetTable>EntityDescription</targetTable>
        <fkColumn>
          <fromColumn>ud_entityDescription_id</fromColumn>
          <targetColumn>ed_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
        <targetTable>ActivityiDescription</targetTable>
        <fkColumn>
          <fromColumn>ud_activityDescription_id</fromColumn>
          <targetColumn>ad_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>WasGeneratedBy</name>
      <description>instance of WasGeneratedBy class</description>
      <column>
        <name>wgb_entity_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasGeneratedBy.entity</utype>  
      </column>
      <column>
        <name>wgb_activity_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasGeneratedBy.activity</utype> 
      </column>
      <column>
        <name>wgb_wasGeneratedByDescription_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasGeneratedBy.wasGeneratedByDescription</utype>
      </column>
      <column>
        <name>wgb_time</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>time.start</ucd>
        <utype>voprov:wasGeneratedBy.time</utype>
      </column>
      <foreignKey>
      <targetTable>WasGeneratedByDescription</targetTable>
        <fkColumn>
          <fromColumn>wgb_wasGeneratedByDescription_id</fromColumn>
          <targetColumn>wgbd_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
      <targetTable>Entity</targetTable>
        <fkColumn>
          <fromColumn>wgb_Entity_id</fromColumn>
	  <targetColumn>e_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
      <targetTable>ActivityDescription</targetTable>
        <fkColumn>
          <fromColumn>wgb_Activity_id</fromColumn>
	  <targetColumn>a_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>WasGeneratedByDescription</name>
      <description>instance of WasGeneratedBy Descriptions</description>
      <column>
        <name>wgbd_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasGeneratedByDescription.id</utype>  
      </column>
      <column>
        <name>wgbd_entityDescription_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasGeneratedByDescription.entityDescription</utype>  
      </column>
      <column>
        <name>wgbd_activityDescription_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasGeneratedByDescription.activityDescription</utype> 
      </column>
      <column>
        <name>wgbd_role</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:WasGeneratedByDescription.role</utype>
      </column>
      <targetTable>EntityDescription</targetTable>
        <fkColumn>
          <fromColumn>wgbd_EntityDescription_id</fromColumn>
	  <targetColumn>ed_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
      <targetTable>ActivityDescription</targetTable>
        <fkColumn>
          <fromColumn>wgbd_ActivityDescription_id</fromColumn>
	  <targetColumn>ad_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>WasAssociatedWith</name>
      <description>instance of WasassociatedWith class</description>
      <column>
        <name>waw_agent_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasAssociatedWith.agent</utype>  
      </column>
      <column>
        <name>waw_activity_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasAssociatedWith.activity</utype> 
      </column>
      <column>
        <name>waw_role</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:WasAssociatedWith.agentRole</utype>
      </column>
      <targetTable>Agent</targetTable>
        <fkColumn>
          <fromColumn>waw_agent_id</fromColumn>
	  <targetColumn>ag_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
      <targetTable>Activity</targetTable>
        <fkColumn>
          <fromColumn>waw_activity_id</fromColumn>
	  <targetColumn>a_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
    <table type="output">
      <name>WasAttributedTo</name>
      <description>instance of WasAttributedTo class</description>
      <column>
        <name>wat_entity_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasAttributedTo.entity</utype>  
      </column>
      <column>
        <name>wat_agent_id</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.id</ucd>
        <utype>voprov:WasAttributedTo.Agent</utype> 
      </column>
      <column>
        <name>wat_role</name>
        <dataType xsi:type="vod:TAPType">VARCHAR</dataType>
        <ucd>meta.code.class</ucd>
        <utype>voprov:WasAttributedTo.agentRole</utype>
      </column>
      <targetTable>Agent</targetTable>
        <fkColumn>
          <fromColumn>wat_entity_id</fromColumn>
	  <targetColumn>e_id</targetColumn>
        </fkColumn>
      </foreignKey>
      <foreignKey>
      <targetTable>Entitty</targetTable>
        <fkColumn>
          <fromColumn>wat_activity_id</fromColumn>
	  <targetColumn>a_id</targetColumn>
        </fkColumn>
      </foreignKey>
    </table>
  </schema>
</vosi:tableset>
\end{lstlisting}

\section{TAP service content exemple. Database dump in VOTable.}
Following exemple gathers information on data collections used or gathered by CDS in the context of Aladin.
It starts from Schmidt plate collections and goes up to HiPS datasets and collections. Theses metadata are organized below into a set of related VOTables.

\begin{lstlisting}[language=XML, caption= TAP schema in xml]

<?xml version="1.0" encoding="utf-8"?>
<!-- Produced with astropy.io.votable version 0.3 http://www.astropy.org/ -->
<VOTABLE version="1.2" xmlns="http://www.ivoa.net/xml/VOTable/v1.2" xmlns:xsi="http://www.w3.org/2001/XMLSchema-instance" xsi:noNamespaceSchemaLocation="http://www.ivoa.net/xml/VOTable/v1.2">
 <RESOURCE type="input data">
  <!-- compatible DM 2018-Jan- 29 -->
<TABLE name="Activity">
   <FIELD ID="a_id" arraysize="*" datatype="char" name="a_id" 
    ucd="meta.id" utype="voprov:Activity.id"/>
   <FIELD ID="a_name_ID" arraysize="*" datatype="char" name="a_name" 
    ucd="meta.title" utype="voprov:Activity.name"/>
   <FIELD ID="a_start_ID" arraysize="*" datatype="char" name="a_startTime" 
    ucd="time.start" utype="voprov:Activity.startTime"/>
   <FIELD ID="a_end_ID" arraysize="*" datatype="char" name="a_endTime" 
    ucd="time.end" utype="voprov:Activity.endTime"/>
   <FIELD ID="a_annotation_ID" arraysize="*" datatype="unicodeChar" 
    name="a_annotation" ucd="meta.description" utype="voprov:Activity.annotation"/>
   <FIELD ID="a_desc_ID" arraysize="*" datatype="char" name="a_description" 
    ucd="meta.id" utype="voprov:Activity.description"/>
   <DATA>
    <TABLEDATA>
  <!-- color composition -->
     <TR>
      <TD>AlaRGB1</TD>
      <TD>Aladin RGB 1</TD>
      <TD>2017-04-18T17:28:00</TD>
      <TD>2017-04-19T17:29:00</TD>
      <TD>Aladin RGB image generation for NGC 6946</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>AlaRGB2</TD>
      <TD>Aladin RGB 2</TD>
      <TD>2017-04-18T17:34:00</TD>
      <TD>2017-04-19T17:35:00</TD>
      <TD>Aladin RGB image generation for Messier 101</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>AlaRGB3</TD>
      <TD>Aladin RGB 3</TD>
      <TD>2017-04-18T17:41:00</TD>
      <TD>2017-04-19T17:42:00</TD>
      <TD>Aladin RGB image generation for Messier 33</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>AlaRGB4</TD>
      <TD>Aladin RGB 4</TD>
      <TD>2017-04-18T17:45:00</TD>
      <TD>2017-04-19T17:46:00</TD>
      <TD>Aladin RGB image generation for Messier 51</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>AlaRGB5</TD>
      <TD>Aladin RGB 5</TD>
      <TD>2017-04-18T17:47:00</TD>
      <TD>2017-04-19T17:48:00</TD>
      <TD>Aladin RGB image generation for Messier 81</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>AlaRGB6</TD>
      <TD>Aladin RGB 6</TD>
      <TD>2017-04-18T17:50:00</TD>
      <TD>2017-04-19T17:51:00</TD>
      <TD>Aladin RGB image generation for Messier 83</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>AlaRGB7</TD>
      <TD>Aladin RGB 7</TD>
      <TD>2017-04-18T17:53:00</TD>
      <TD>2017-04-19T17:54:00</TD>
      <TD>Aladin RGB image generation for Messier 87</TD>
      <TD>AlaRGB</TD>
     </TR>
     <TR>
      <TD>stsciNum-21</TD>
      <TD>Num DSS2 POSSII 061 J</TD>
      <TD>2006-06-29T15:32:50</TD>
      <TD/>
      <TD>DSS2 Digitization of plates at Stsci POSSII 061 J</TD>
      <TD>stsciNum</TD>
     </TR>
   <!--  numerisation -->
     <TR>
      <TD>stsciNum-22</TD>
      <TD>Num DSS2 POSSII 061 F</TD>
      <TD>2006-04-04T16:39:18</TD>
      <TD/>
      <TD>DSS2 Digitization of plates at Stsci POSSII 061 F</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-23</TD>
      <TD>Num DSS2 POSSII 061 N</TD>
      <TD>2006-04-04T16:52:00</TD>
      <TD/>
      <TD>DSS2 Digitization of plates at Stsci POSSII 061 N</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-24</TD>
      <TD>Num DSS2 POSSII 143 J</TD>
      <TD>2006-04-04T16:10:12</TD>
      <TD>2006-04-04T16:10:30</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 143 J</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-25</TD>
      <TD>Num DSS2 POSSII 143 F</TD>
      <TD>2006-04-04T16:25:45</TD>
      <TD>2006-04-04T16:25:55</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 143 F</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-26</TD>
      <TD>Num DSS2 POSSII 143 N</TD>
      <TD>2006-04-04T16:31:01</TD>
      <TD>2006-04-04T16:31:21</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 143 N</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-27</TD>
      <TD>Num DSS2 POSSII 270 J</TD>
      <TD>2006-04-04T16:17:26</TD>
      <TD>2006-04-04T16:17:36</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 270 J</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-28</TD>
      <TD>Num DSS2 POSSII 270 F</TD>
      <TD>2006-04-04T16:17:27</TD>
      <TD>2006-04-04T16:17:37</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 270 F</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-29</TD>
      <TD>Num DSS2 POSSII 270 N</TD>
      <TD>2006-04-04T16:35:11</TD>
      <TD>2006-04-04T16:35:21</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 270 N</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2a</TD>
      <TD>Num DSS2 POSSII 174 J</TD>
      <TD>2006-04-04T16:36:03</TD>
      <TD>2006-04-04T16:36:13</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 174 J</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2b</TD>
      <TD>Num DSS2 POSSII 174 F</TD>
      <TD>2006-04-04T16:22:28</TD>
      <TD>2006-04-04T16:22:38</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 174 F</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2c</TD>
      <TD>Num DSS2 POSSII 175 N</TD>
      <TD>2006-04-04T16:41:00</TD>
      <TD>2006-04-04T16:41:10</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 175 N</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2d</TD>
      <TD>Num DSS2 POSSII 413 J</TD>
      <TD>2006-04-04T16:19:43</TD>
      <TD>2006-04-04T16:19:53</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 413 J</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2e</TD>
      <TD>Num DSS2 POSSII 413 F</TD>
      <TD>2006-04-04T16:18:05</TD>
      <TD>2006-04-04T16:18:15</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 413 F</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2f</TD>
      <TD>Num DSS2 POSSII 413 N</TD>
      <TD>2006-06-29T15:32:42</TD>
      <TD>2006-06-29T15:32:52</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 413 N</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2g</TD>
      <TD>Num DSS2 POSSII 644 J</TD>
      <TD>2006-04-04T16:07:36</TD>
      <TD>2006-04-04T16:07:46</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 644 J</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2h</TD>
      <TD>Num DSS2 POSSII 644 F</TD>
      <TD>2006-04-04T16:11:58</TD>
      <TD>2006-04-04T16:12:08</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 644 F</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2i</TD>
      <TD>Num DSS2 POSSII 644 N</TD>
      <TD>2006-04-04T16:11:58</TD>
      <TD>2006-04-04T16:11:58</TD>
      <TD>DSS2 Digitization of plates at Stsci POSSII 644 N</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>stsciNum-2j</TD>
      <TD>Num DSS2 SERC 445 I</TD>
      <TD>2006-04-04T16:36:09</TD>
      <TD>2006-04-04T16:36:19</TD>
      <TD>DSS2 Digitization of plates at Stsci SERC 445 I</TD>
      <TD>stsciNum</TD>
     </TR>
     <TR>
      <TD>MAMANUM-1</TD>
      <TD>numerisation mama SERC J 444</TD>
      <TD>1994-02-04T09:00:00</TD>
      <TD>1994-02-04T14:57:00</TD>
      <TD>GEPI MAMA Digitization of plate SERC 444J</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
     <TR>
      <TD>MAMANUM-2</TD>
      <TD>numerisation mama ESO R 444</TD>
      <TD>1994-01-28T09:03:00</TD>
      <TD>1994-01-28T13:07:00</TD>
      <TD>GEPI MAMA Digitization of plate ESO 444R</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
     <TR>
      <TD>MAMANUM-3</TD>
      <TD>numerisation mama ESO R 445</TD>
      <TD>1993-08-12T09:17:00</TD>
      <TD>1993-08-12T13:32:00</TD>
      <TD>GEPI MAMA Digitization of plate ESO 445R</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
     <TR>
      <TD>MAMANUM-4</TD>
      <TD>numerisation mama ESO R 446</TD>
      <TD>1993-08-13T09:13:00</TD>
      <TD>1993-08-13T13:25:00</TD>
      <TD>GEPI MAMA Digitization of plate ESO 446R</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
     <TR>
      <TD>MAMANUM-5</TD>
      <TD>numerisation mama ESO R 447</TD>
      <TD>1993-08-14T08:58:00</TD>
      <TD>1993-08-14T13:00:00</TD>
      <TD>GEPI MAMA Digitization of plate ESO 447R</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
     <TR>
      <TD>MAMANUM-ESOR</TD>
      <TD>numerisation mama ESO survey</TD>
      <TD>1993-08-01T08:58:00</TD>
      <TD>1993-08-31T13:00:04</TD>
      <TD>GEPI MAMA Digitization of  ESO plates</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
     <TR>
      <TD>MAMANUM-SERCJ</TD>
      <TD>numerisation mama SERC survey</TD>
      <TD>1994-02-02T08:58:00</TD>
      <TD>1994-03-04T13:00:00</TD>
      <TD>GEPI MAMA Digitization of SERC plate</TD>
      <TD>MAMANUM-xyz</TD>
     </TR>
   <!--cut-out activity -->
     <TR>
      <TD>cds_cutoutj061</TD>
      <TD>Cut out Aladin POSSII 061 J</TD>
      <TD>2017-04-18T16:33:00</TD>
      <TD>2017-04-19T16:34:00</TD>
      <TD>Cut out CDS- soda service POSSII 061 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutf061</TD>
      <TD>Cut out Aladin POSSII 061 F</TD>
      <TD>2017-04-18T16:34:00</TD>
      <TD>2017-04-19T16:35:00</TD>
      <TD>Cut out CDS- soda service POSSII 061 F</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutn061</TD>
      <TD>Cut out Aladin POSSII 061 N</TD>
      <TD>2017-04-18T16:35:00</TD>
      <TD>2017-04-19T16:36:00</TD>
      <TD>Cut out CDS- soda service POSSII 061 N</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutj143</TD>
      <TD>Cut out Aladin POSSII 143 J</TD>
      <TD>2017-04-18T16:36:00</TD>
      <TD>2017-04-19T16:37:00</TD>
      <TD>Cut out CDS- soda service POSSII 143 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutf143</TD>
      <TD>Cut out Aladin POSSII 143 F</TD>
      <TD>2017-04-18T16:37:00</TD>
      <TD>2017-04-19T16:38:00</TD>
      <TD>Cut out CDS- soda service POSSII 143 F</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutn143</TD>
      <TD>Cut out Aladin POSSII 143 N</TD>
      <TD>2017-04-18T16:38:00</TD>
      <TD>2017-04-19T16:39:00</TD>
      <TD>Cut out CDS- soda service POSSII 143 N</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutj174</TD>
      <TD>Cut out Aladin POSSII 174 J</TD>
      <TD>2017-04-18T16:39:00</TD>
      <TD>2017-04-19T16:40:00</TD>
      <TD>Cut out CDS- soda service POSSII 174 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutf174</TD>
      <TD>Cut out Aladin POSSII 174 F</TD>
      <TD>2017-04-18T16:40:00</TD>
      <TD>2017-04-19T16:41:00</TD>
      <TD>Cut out CDS- soda service POSSII 174 F</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutn175</TD>
      <TD>Cut out Aladin POSSII 175 N</TD>
      <TD>2017-04-18T16:41:00</TD>
      <TD>2017-04-19T16:42:00</TD>
      <TD>Cut out CDS- soda service POSSII 175 N</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutj270</TD>
      <TD>Cut out Aladin POSSII 270 J</TD>
      <TD>2017-04-18T16:42:00</TD>
      <TD>2017-04-19T16:42:30</TD>
      <TD>Cut out CDS- soda service POSSII 270 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutf270</TD>
      <TD>Cut out Aladin POSSII 270 F</TD>
      <TD>2017-04-18T16:43:00</TD>
      <TD>2017-04-19T16:43:30</TD>
      <TD>Cut out CDS- soda service POSSII 270 F</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutn270</TD>
      <TD>Cut out Aladin POSSII 270 N</TD>
      <TD>2017-04-18T16:44:00</TD>
      <TD>2017-04-19T16:44:30</TD>
      <TD>Cut out CDS- soda service POSSII 270 N</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutj413</TD>
      <TD>Cut out Aladin POSSII 143 J</TD>
      <TD>2017-04-18T16:45:00</TD>
      <TD>2017-04-19T16:45:30</TD>
      <TD>Cut out CDS- soda service POSSII 143 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutf413</TD>
      <TD>Cut out Aladin POSSII 143 F</TD>
      <TD>2017-04-18T16:46:00</TD>
      <TD>2017-04-19T16:46:30</TD>
      <TD>Cut out CDS- soda service POSSII 143 F</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutn413</TD>
      <TD>Cut out Aladin POSSII 143 N</TD>
      <TD>2017-04-18T16:47:00</TD>
      <TD>2017-04-19T16:47:40</TD>
      <TD>Cut out CDS- soda service POSSII 143 N</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutj644</TD>
      <TD>Cut out Aladin POSSII 644 J</TD>
      <TD>2017-04-18T16:48:00</TD>
      <TD>2017-04-19T16:48:25</TD>
      <TD>Cut out CDS- soda service POSSII 644 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutf644</TD>
      <TD>Cut out Aladin POSSII 644 F</TD>
      <TD>2017-04-18T16:49:00</TD>
      <TD>2017-04-19T16:49:26</TD>
      <TD>Cut out CDS- soda service POSSII 644 F</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutn644</TD>
      <TD>Cut out Aladin POSSII 644 N</TD>
      <TD>2017-04-18T16:50:00</TD>
      <TD>2017-04-19T16:50:30</TD>
      <TD>Cut out CDS- soda service POSSII 644 N</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutouti445</TD>
      <TD>Cut out Aladin SERC 445 I</TD>
      <TD>2017-04-18T16:52:00</TD>
      <TD>2017-04-19T16:52:20</TD>
      <TD>Cut out CDS- soda service SERC 445 I</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutj444</TD>
      <TD>Cut out Aladin SERC 444 J</TD>
      <TD>2017-04-18T16:54:00</TD>
      <TD>2017-04-19T16:54:20</TD>
      <TD>Cut out CDS- soda service SERC 444 J</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>cds_cutoutr444</TD>
      <TD>Cut out Aladin ESO 444 R</TD>
      <TD>2017-04-18T16:55:00</TD>
      <TD>2017-04-19T16:55:30</TD>
      <TD>Cut out CDS- soda service ESO 444 R</TD>
      <TD>cds_cutout</TD>
     </TR>
     <TR>
      <TD>EHG1</TD>
      <TD>ESO HiPS generation 1</TD>
      <TD>2016-07-18T09:45:00</TD>
      <TD>2016-07-20T10:00:00</TD>
      <TD>
       this activity is final generation of HiPS for ESO Mama survey
      </TD>
      <TD>HipsgenM</TD>
     </TR>
     <TR>
      <TD>EHG2</TD>
      <TD>SERC HiPS generation 1</TD>
      <TD>2016-07-11T10:45:00</TD>
      <TD>2016-07-14T03:07:00</TD>
      <TD>
       this activity is final generation of HiPS for SERC Mama survey
      </TD>
      <TD>HipsgenM</TD>
     </TR>
    </TABLEDATA>
   </DATA>
</TABLE>
<TABLE name="activity descriptions" utype="voprov:ActivityDesc" >
     <FIELD name="ad_id" ucd="meta.id" utype ="voprov:ActivityDescription.id" datatype="char" arraysize="*"/>
     <FIELD name="ad_name" ucd="meta.title" utype="voprov:ActivityDescription.name" 
       datatype="char" arraysize="*"/>
     <FIELD name="ad_type" ucd="meta.code.class" utype="voprov:ActivityDescription.type" 
        datatype="char" arraysize="*" />
     <FIELD name ="ad_subtype" ucd="meta.code.class" utype="voprov:ActivityDescription.subtype"  
       datatype="char" arraysize="*"/>
     <FIELD name="ad_doculink" ucd="meta.ref.url" utype="voprov:ActivityDescription.doculink" 
       datatype="char" arraysize="*"  />
     <FIELD name="ad_annotation" ucd="meta.description" utype="voprov:ActivityDescription.annotation" 
       datatype="unicodeChar" arraysize="*"  />
  <DATA>
    <TABLEDATA>
        <TR>
            <TD>AlaRGB</TD>
            <TD>Aladin RGB image generation algorithm</TD>
            <TD>RGBencoding</TD>
            <TD></TD>
            <TD>http://cds.u-strasbg.fr/aladin.gml</TD>
            <TD></TD>
        </TR>
        <TR>
            <TD>stsciNum</TD>
            <TD>STSCI scan</TD>
            <TD>DSS2 Plate digitization</TD>
            <TD>Micro densitometer</TD>
            <TD>http://stsci/DSS2/scan.html</TD>
            <TD></TD>
        </TR>
        <TR>
            <TD>MAMANUM-xyz</TD>
            <TD>MAMA digitizer pipeline</TD>
            <TD>Plate digitization</TD>
            <TD>Micro densitometer</TD>
            <TD>http://gepi/mama/scan.html</TD>
            <TD></TD>
        </TR>
        <TR>
            <TD>cds_cutout</TD>
            <TD>cut out service </TD>
            <TD>soda</TD>
            <TD>cutout</TD>
            <TD>http://cds.u-strasbg.fr/aladin-soda.gml</TD>
            <TD></TD>
        </TR>
        <TR>
            <TD>HipsgenM</TD>
            <TD>HiPS Generation MEAN</TD>
            <TD>HiPSgen</TD>
            <TD>HIPSgen_MEAN</TD>
            <TD>http://cds.u-strasbg.fr/HiPSGEN-Documentation.html</TD>
            <TD></TD>
        </TR>
    </TABLEDATA>
  </DATA>
</TABLE>
<!-- entity table -->
<TABLE name="entities" utype="voprov:Entity" >
       <FIELD name="e_id" ucd="meta.id" utype="voprov:Entity.id" datatype = "char" arraysize="*" />
      <FIELD name="e_name" ucd="meta.title" utype="voprov:Entity.name" datatype="char" arraysize="*"  />
      <FIELD name="e_type" ucd="meta.code.class" utype="voprov:Entity.type" datatype="char" arraysize="*" />
      <FIELD name="e_annotation" ucd="meta.description" utype="voprov:Entity.annotation" datatype="unicodeChar" arraysize="*" />
      <FIELD name="e_rights" ucd="meta.code.class" utype="voprov:Entity.rights" datatype="char" arraysize="*" />
      <FIELD name="e_description" ucd="meta.id" utype="voprov:Entity.description" datatype="char" arraysize="*" /> 
  <DATA>
    <TABLEDATA>
#e_rights devrait etre dans obscore losrsque le dataset est d\'ecrit . 
#ce droit est attach\'e au fichier , pas a l'entit\'e ? 
<!-- 7 color images -->
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_NGC6946</TD>
            <TD>RGB DSS2 image for NGC 6946</TD>
            <TD></TD>
            <TD>
           This is a PNG RGB image built from DSS2 with Aladin for galaxy NGC 6946
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M101</TD>
            <TD>RGB DSS2 image for Messier 101</TD>
            <TD></TD>
            <TD>
            This is a PNG RGB image built from DSS2 with Aladin for galaxy Messier 101
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M33</TD>
            <TD>RGB DSS2 image for Messier 33</TD>
            <TD></TD>
            <TD>
            This is a PNG RGB image built from DSS2 with Aladin for galaxy Messier 33
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M51</TD>
            <TD>RGB DSS2 image for Messier 51</TD>
            <TD></TD>
            <TD>
             This is a PNG RGB image built from DSS2 with Aladin for galaxy Messier 51
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M81</TD>
            <TD>RGB DSS2 image for Messier 81</TD>
            <TD></TD>
            <TD>
            This is a PNG RGB image built from DSS2 with Aladin for galaxy Messier 81
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M83</TD>
            <TD>RGB DSS2 image for Messier 83</TD>
            <TD></TD>
            <TD>
            This is a PNG RGB image built from DSS2 with Aladin for galaxy Messier 83
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M87</TD>
            <TD>RGB DSS2 image for Messier 87</TD>
            <TD></TD>
            <TD>
             This is a PNG RGB image built from DSS2 with Aladin for galaxy Messier 87
            </TD>
            <TD></TD>
            <TD>color#RGB</TD>
        </TR>
<!--! 21 cutouts -->
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.061</TD>
            <TD>POSSII Infra Red Survey DSS2 M81</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the POSSII Schmidt survey around Messier 81
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.061</TD>
            <TD>POSSII Blue Survey DSS2 M81</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey around Messier 81
             </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.061</TD>
            <TD>POSSII Red Survey DSS2 M81</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey around Messier 81
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.143</TD>
            <TD>POSSII Blue Survey DSS2 NGC6946</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Blue POSSII Schmidt survey around NGC 6946
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.143</TD>
            <TD>POSSII Red Survey DSS2 NGC6946</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Red POSSII Schmidt survey around NGC 6946
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.143</TD>
            <TD>POSSII Infra Red Survey DSS2 NGC6946</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Infra red POSSII Schmidt survey around  NGC 6946
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.174</TD>
            <TD>POSSII Blue Survey DSS2 M101</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Blue POSSII Schmidt survey around Messier 101
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.174</TD>
            <TD>POSSII Red Survey DSS2 M101</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Red POSSII Schmidt survey around Messier 101
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.175</TD>
            <TD>POSSII Infra Red Survey DSS2 M101</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Infra red POSSII Schmidt survey around Messier 101
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.270</TD>
            <TD>POSSII Blue Survey DSS2 M51</TD>
            <TD></TD>
            <TD>This is the DSS2 digitization of the Blue POSSII Schmidt survey around Messier 51</TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.270</TD>
            <TD>POSSII Red Survey DSS2 M51</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey around Messier 51
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.270</TD>
            <TD>POSSII Infra Red Survey DSS2 M51</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Infra red POSSII Schmidt survey around Messier 51
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.413</TD>
            <TD>POSSII Blue Survey DSS2 M33</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey around Messier 33
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.413</TD>
            <TD>POSSII Red Survey DSS2 M33</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey around Messier 33
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.413</TD>
            <TD>POSSII Infra Red Survey DSS2 M33</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Infra red POSSII Schmidt survey around Messier 33
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.644</TD>
            <TD>POSSII Blue Survey DSS2 M87</TD>
            <TD></TD>
            <TD>
            This is the cut-out DSS2 digitization of the Blue POSSII Schmidt survey around Messier 87
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.644</TD>
            <TD>POSSII Red Survey DSS2 M87</TD>
            <TD></TD>
            <TD>
            This is the cut-out DSS2 digitization of the Red POSSII Schmidt survey around Messier 87
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.644</TD>
            <TD>POSSII Infra Red Survey DSS2 M87</TD>
            <TD></TD>
            <TD>
            This is the cut-out DSS2 digitization of the Infra red POSSII Schmidt survey around Messier 87
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/DSS2/SERC#SERC.I-DSS2.445</TD>
            <TD>SERC Infra Red Survey DSS2 M83</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Infra REd SERC Schmidt survey around Messier 83
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/MAMA/SERC#SERC.J-MAMA.444</TD>
            <TD>SERC  Blue Survey MAMA M83</TD>
            <TD></TD>
            <TD>
             This is the MAMA digitization of the blue SERC Schmidt survey around Messier 83
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/MAMA/ESO#ESO.R-MAMA.444</TD>
            <TD>ESO Infra Red Survey MAMA M83</TD>
            <TD></TD>
            <TD>
            This is the MAMA digitization of the Red MAMA Schmidt survey around Messier 83
            </TD>
            <TD></TD>
            <TD>cutout#DSS2_MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.061</TD>
            <TD>POSSII Blue Survey DSS2 061</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey plate 061
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.061</TD>
            <TD>POSSII Red Survey DSS2 061</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Red POSSII Schmidt survey around plate 061
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.061</TD>
            <TD>POSSII Infra Red Survey DSS2 061</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Infra Red POSSII Schmidt survey plate  061
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.143</TD>
            <TD>POSSII Blue Survey DSS2 143</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Blue POSSII Schmidt survey around plate 143
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.143</TD>
            <TD>POSSII Red Survey DSS2 143</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey around plate 143
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.143</TD>
            <TD>POSSII Infra Red Survey DSS2 143</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 143
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.174</TD>
            <TD>POSSII Blue Survey DSS2 174</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey around plate 174
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.174</TD>
            <TD>POSSII Red Survey DSS2 174</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Red POSSII Schmidt survey plate 174
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.175</TD>
            <TD>POSSII Infra Red Survey DSS2 M101</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 175
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.270</TD>
            <TD>POSSII Blue Survey DSS2 270</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey plate 270
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.270</TD>
            <TD>POSSII Red Survey DSS2 270</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey plate 270
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.270</TD>
            <TD>POSSII Infra Red Survey DSS2 270</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 270
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.413</TD>
            <TD>POSSII Blue Survey DSS2 413</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Blue POSSII Schmidt survey plate 413
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.413</TD>
            <TD>POSSII Red Survey DSS2 413</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Red POSSII Schmidt survey plate 413
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.413</TD>
            <TD>POSSII Infra Red Survey DSS2 413</TD>
            <TD></TD>
            <TD>
            This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 413
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.644</TD>
            <TD>Digital POSSII plate Blue Survey DSS2 644</TD>
            <TD></TD>
            <TD>
            This is the numerical plate of the Blue POSSII Schmidt survey 
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.644</TD>
            <TD>Digital POSSII plate Red Survey DSS2 644</TD>
            <TD></TD>
            <TD>
            This is the numerical plate of the Red POSSII Schmidt survey around Messier 87
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.644</TD>
            <TD>Digital POSSII plate Infra Red Survey DSS2 644</TD>
            <TD></TD>
            <TD>
             This is the numerical  plate of the Infra red POSSII Schmidt survey around Messier 87
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://STSci/Num#SERC.I-DSS2.445</TD>
            <TD>Digital SERC plate Infra Red Survey DSS2 445</TD>
            <TD></TD>
            <TD>
             This is the numerical  plate of the Infra red SERC Schmidt survey for plate 445
            </TD>
            <TD></TD>
            <TD>num#DSS2</TD>
        </TR>
        <TR>
            <TD>ivo://gepi/MAMA/Num#SERC.J-MAMA.444</TD>
            <TD>SERC J Survey MAMA plate 444</TD>
            <TD></TD>
            <TD>
             This is the MAMA digitization of the blue SERC  Schmidt plate 444
            </TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.444</TD>
            <TD>ESO  Red Survey MAMA plate 444</TD>
            <TD></TD>
            <TD>
             This is the MAMA digitization of the Red MAMA Schmidt plate 444 
            </TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.445</TD>
            <TD>ESO Red Survey MAMA plate 445</TD>
            <TD></TD>
            <TD>
             This is the MAMA digitization of the Red MAMA Schmidt plate 445</TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.446</TD>
            <TD>ESO Red Survey MAMA plate 446</TD>
            <TD></TD>
            <TD>
            This is the MAMA digitization of the Red MAMA Schmidt plate 446
            </TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.447</TD>
            <TD>ESO Red Survey MAMA plate 447</TD>
            <TD></TD>
            <TD>
             This is the MAMA digitization of the Red MAMA Schmidt plate 447
             </TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
<!--! Schmidt photo plates -->
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.644</TD>
            <TD>POSSII plate Blue Survey DSS2 plate 644</TD>
            <TD></TD>
            <TD>
              This is the plate of the Blue POSSII Schmidt survey plate 644
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.644</TD>
            <TD>POSSII plate Red Survey DSS2  plate 644</TD>
            <TD></TD>
            <TD>
              This is the plate of the Red POSSII Schmidt survey plate 644
             </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.644</TD>
            <TD>POSSII plate Infra Red Survey DSS2  plate 644</TD>
            <TD></TD>
            <TD>
             This is plate of the Infra red POSSII Schmidt survey plate 644
             </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.061</TD>
            <TD>POSSII Infra Red Survey DSS2 plate 644</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Infra Red POSSII Schmidt survey plate  061
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.061</TD>
            <TD>POSSII Blue Survey DSS2 plate 061</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Blue POSSII Schmidt survey plate 061
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.061</TD>
            <TD>POSSII Red Survey DSS2 061</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Red POSSII Schmidt survey around plate 061
             </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.143</TD>
            <TD>POSSII Blue Survey DSS2 plate 143</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey around plate 143
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.143</TD>
            <TD>POSSII Red Survey DSS2 plate 143</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey around plate 143
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.143</TD>
            <TD>POSSII Infra Red Survey DSS2 plate 143</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 143
             </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.174</TD>
            <TD>POSSII Blue Survey DSS2 plate 174</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Blue POSSII Schmidt survey  plate 174
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.174</TD>
            <TD>POSSII Red Survey DSS2 plate 174</TD>
            <TD></TD>
            <TD>
             This is the DSS2 digitization of the Red POSSII Schmidt survey plate 174
             </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.175</TD>
            <TD>POSSII Infra Red Survey DSS2 M101</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 175
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.270</TD>
            <TD>POSSII Blue Survey DSS2 270</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Blue POSSII Schmidt survey plate 270
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.270</TD>
            <TD>POSSII Red Survey DSS2 270</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Red POSSII Schmidt survey plate 270
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.270</TD>
            <TD>POSSII Infra Red Survey DSS2 270</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 270
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.413</TD>
            <TD>POSSII Blue Survey DSS2 413</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Blue POSSII Schmidt survey plate 413
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.413</TD>
            <TD>POSSII Red Survey DSS2 413</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Red POSSII Schmidt survey plate 413
             </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.413</TD>
            <TD>POSSII Infra Red Survey DSS2 413</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 413
            </TD>
            <TD></TD>
            <TD>Plate#POSSII</TD>
        </TR>
        <TR>
            <TD>ivo://SERC/Plate#SERC.I.445</TD>
            <TD>SERC Infra Red Survey  445</TD>
            <TD></TD>
            <TD>
              This is   the Infra red SERC Schmidt survey plate 445
            </TD>
            <TD></TD>
            <TD>Plate#SERC</TD>
        </TR>
        <TR>
            <TD>ivo://SERC/Plate#SERC.J.444</TD>
            <TD>SERC blue Survey  444</TD>
            <TD></TD>
            <TD>
               This is   the blue SERC Schmidt survey plate 444
            </TD>
            <TD></TD>
            <TD>Plate#SERC</TD>
        </TR>
        <TR>
            <TD>ivo://ESO/Plate#ESO.R.444</TD>
            <TD>ESO  Red Survey  444</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 413
            </TD>
            <TD></TD>
            <TD>Plate#ESO</TD>
        </TR>
        <TR>
            <TD>ivo://ESO/Plate#ESO.R.445</TD>
            <TD>ESO  Red Survey  444</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 445
            </TD>
            <TD></TD>
            <TD>Plate#ESO</TD>
        </TR>
        <TR>
            <TD>ivo://ESO/Plate#ESO.R.446</TD>
            <TD>ESO  Red Survey  445</TD>
            <TD></TD>
            <TD>
              This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 446
            </TD>
            <TD></TD>
            <TD>Plate#ESO</TD>
        </TR>
        <TR>
            <TD>ivo://ESO/Plate#ESO.R.447</TD>
            <TD>ESO  Red Survey  446</TD>
            <TD></TD>
            <TD>
                This is the DSS2 digitization of the Infra red POSSII Schmidt survey plate 447
            </TD>
            <TD></TD>
            <TD>Plate#ESO</TD>
        </TR>
<!--! ESO HiPS -->
        <TR>
            <TD>ivo://cds/P/MAMA/ESO-R</TD>
            <TD>ESO-R MAMA HiPS at CDS</TD>
            <TD></TD>
            <TD>
              This is the HiPS version of ESO Schmidt survey digitized by Mama and processed by CDS
            </TD>
            <TD></TD>
            <TD>HipS#CDS</TD>
        </TR>
        <TR>
            <TD>ivo://obspm/gepi/ESO-MAMA</TD>
            <TD>ESO-R Survey Mama GEPI</TD>
            <TD></TD>
            <TD>
              This is the ESO Schmidt survey collection digitized by MAMA at CAI curated by GEPI
            </TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
        <TR>
            <TD>ivo://cds/P/MAMA/SERC-J</TD>
            <TD>SERC J MAMA HiPS at CDS</TD>
            <TD></TD>
            <TD>
              This is the HiPS version of SERC J Schmidt survey digitized by Mama and processed by CDS 
            </TD>
            <TD></TD>
            <TD>HipS#CDS</TD>
        </TR>
        <TR>
            <TD>ivo://obspm/gepi/SERCJ-MAMA</TD>
            <TD>SERC J Survey Mama GEPI</TD>
            <TD></TD>
            <TD>
              This is the SERC J  Schmidt survey collection digitized by MAMA at CAI curated by GEPI
            </TD>
            <TD></TD>
            <TD>num#MAMA</TD>
        </TR>
    </TABLEDATA>
  </DATA>
</TABLE>
<TABLE name="entityDescription" utype="voprov:entityDescription" >
    <FIELD name="ed_id" ucd="meta.id" utype="voprov:EntityDescription.id" 
      datatype = "char" arraysize="*" />
    <FIELD name="ed_name" ucd="meta.title" utype="voprov:EntityDescription.name" 
      datatype="char" arraysize="*"  />
    <FIELD name="ed_category" ucd="meta.code.class" utype="voprov:EntityDescription.category" 
      datatype="char" arraysize="*" />
    <FIELD name="ed_annotation" ucd="meta.description" utype="voprov:EntityDescription.annotation" 
      datatype="unicodeChar" arraysize="*" />
    <FIELD name="ed_doculink" ucd="meta.ref.url" utype="voprov:EntityDescription.doculink" 
      datatype="char" arraysize="*"   /> 
	   
<DATA>
<TABLEDATA>
<!--  color images -->
         <TR>
            <TD>color#RGB</TD>
            <TD>RGB image description</TD>
            <TD></TD>
            <TD>The images sharing this description  are  PNG RGB images built 
             from multiband cutouts with Aladin </TD>
            <TD>http://cds.u-strasbg.fr/aladinRGB.gml</TD>
         </TR>
<!--  cutouts -->
         <TR>
            <TD>cutout#DSS2_MAMA</TD>
            <TD></TD>
            <TD></TD>
            <TD>The images sharing this description have been extracted by cutout
            from Aladin server </TD>
            <TD>http://cds.u-strasbg.fr/aladinCutouts.gml</TD>
         </TR>
<!-- digitized plates -->
         <TR>
            <TD>Num#DSS2</TD>
            <TD>digitized DSS2 plates</TD>
            <TD></TD>
            <TD>This is the description of DSS2 digitization </TD>
            <TD>http://stsci.edu/dss-plates.html</TD>
         </TR>
         <TR>
            <TD>Num#MAMA</TD>
            <TD>digitized MAMA plates</TD>
            <TD></TD>
            <TD>This is the description of  MAMA digitization </TD>
            <TD>http://stsci.edu/mama-plates.html</TD>
         </TR>
<!-- Schmidt photographic plates -->
         <TR>
            <TD>Plate#POSSII</TD>
            <TD>plates POSSII  Survey </TD>
            <TD></TD>
            <TD>This is the description of plates from the POSSII Schmidt survey</TD>
            <TD>http://palomar-telescope.edu/possii-plates.html</TD>
         </TR>
         <TR>
            <TD>Plate#SERC</TD>
            <TD>plates SERC  Survey </TD>
            <TD></TD>
            <TD>This is the description of plates from the SERC Schmidt survey</TD>
            <TD>http://serc-telescope.edu/serc-plates.html</TD>
         </TR>
         <TR>
            <TD>Plate#ESO</TD>
            <TD>plates ESO  Survey </TD>
            <TD></TD>
            <TD>This is the description of plates from the ESO Schmidt survey</TD>
            <TD>http://eso-telescope.edu/eso-plates.html</TD>
         </TR>
<!--  ESO HiPS -->
         <TR>
            <TD>HipS#CDS</TD>
            <TD>HiPS at CDS</TD>
            <TD></TD>
            <TD>This is the description of HiPS operated by CDS</TD>
            <TD>http://cds.u-strasbg.fr/hips/documentation.html#structure</TD>
         </TR>
</TABLEDATA>
</DATA>
</TABLE>
<!-- agent table -->
<!-- table agent compatible DM 2018-Jan- 29 -->
  <TABLE name="Agent">
   <FIELD ID="agt_id" arraysize="*" datatype="unicodeChar" name="ag_id" ucd="meta.id" utype="voprov:Agent.id"/>
   <FIELD ID="agt_name" arraysize="*" datatype="unicodeChar" name="ag_name" ucd="meta.title" utype="voprov:Agent.name"/>
   <FIELD ID="agt_type" arraysize="*" datatype="unicodeChar" name="ag_type" ucd="meta.code.class" utype="voprov:Agent.type"/>
   <FIELD ID="agt_email" arraysize="*" datatype="unicodeChar" name="ag_email" ucd="meta.email" utype="voprov:Agent.email"/>
   <FIELD ID="agt_phone" arraysize="*" datatype="unicodeChar" name="ag_phone" ucd="meta.phone" utype="voprov:Agent.phone"/>
   <FIELD ID="agt_adress" arraysize="*" datatype="unicodeChar" name="ag_adress" ucd="meta.address" utype="voprov:Agent.address"/>
   <!-- meta.adress  n'existe pas en ucd  --> 
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>ivo://cds</TD>
      <TD>Centre de donnees astronomiques de Strasbourg</TD>
      <TD>Organisation</TD>
      <TD>cdsquestion@astro.unistra.fr</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>bonnarel</TD>
      <TD>Francois Bonnarel</TD>
      <TD>Individual</TD>
      <TD>francois.bonnarel@astro.unistra.fr</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>ivo://stsci</TD>
      <TD>Space Telescope</TD>
      <TD>Organisation</TD>
      <TD>contact@stsci.edu</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>ivo://Palomar</TD>
      <TD>Palomar Schmidt Telescope</TD>
      <TD>Organisation</TD>
      <TD>contact@palomar-telescope.edu</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>ivo://ESO</TD>
      <TD>ESO Schmidt Telescope</TD>
      <TD>Organisation</TD>
      <TD>contact@eso.org</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>ivo://SERC</TD>
      <TD>SERC Siding Spring Schmidt Telescope</TD>
      <TD>Organisation</TD>
      <TD>contact@serc.au</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>ivo://gepi</TD>
      <TD>GEPI</TD>
      <TD>Organisation</TD>
      <TD>lesidaner@obspm.fr</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>ivo://obspm/cai</TD>
      <TD>centre analyse des images</TD>
      <TD>Organisation</TD>
      <TD>lesidaner@obspm.fr</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>buga</TD>
      <TD>Mihaela Buga</TD>
      <TD>Individual</TD>
      <TD>buga@astro.unistra.fr</TD>
      <TD/>
      <TD/>
    </TR>
     <TR>
      <TD>Guibert</TD>
      <TD>Jean Guibert</TD>
      <TD>Individual</TD>
      <TD>guibert@obspm.fr</TD>
      <TD/>
      <TD/>
    </TR>
    </TABLEDATA>
   </DATA>
  </TABLE>
<!-- relation tables -->
  <TABLE name="WasAttributedTo">
  <!-- here is the dump from the CDS test DB on 2018-Jan- 29 
  relation between an Agent and an Entity -->
  <!-- compatible DM 2018-Jan- 29 -->
   <FIELD ID="att_idacribedentity" arraysize="*" datatype="char" name="wat_entity_id" 
    ucd="meta.id" utype="voprov:WasAttributedTo.entity"/>
   <FIELD ID="att_idacribedagent" arraysize="*" datatype="unicodeChar" name="wat_agent_id" ucd="meta.id" utype="voprov:WasAttributedTo.agent"/>
   <FIELD ID="att_role" arraysize="*" datatype="char" name="wat_role" ucd="meta.code.class" utype="voprov:WasAttributedTo.agentRole"/>
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_NGC6946</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_NGC6946</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M101</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M101</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M33</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M33</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M51</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M51</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M81</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M81</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M83</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M83</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M87</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M87</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.644</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.644</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.644</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.061</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.061</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.061</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.143</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.143</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.143</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.270</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.270</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.270</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.174</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.174</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.175</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.413</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.413</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.413</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/SERC#SERC.I-DSS2.445</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/SERC#SERC.J-MAMA.444</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/ESO#ESO.R-MAMA.444</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.644</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.644</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.644</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.061</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.061</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.061</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.143</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.143</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.143</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.174</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.174</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.175</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.413</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.413</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.413</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.270</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.270</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.270</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#SERC.I-DSS2.445</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#SERC.J-MAMA.444</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.444</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.J.644</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.F.644</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.N.644</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.J.061</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.F.061</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.N.061</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.J.143</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.F.143</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.N.143</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.J.174</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.F.174</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.N.175</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.J.413</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.F.413</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.N.413</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.J.270</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.F.270</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://POSSII/Plate#POSSII.N.270</TD>
      <TD>ivo://Palomar</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://SERC/Plate#SERC.I.445</TD>
      <TD>ivo://SERC</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://SERC/Plate#SERC.J.444</TD>
      <TD>ivo://SERC</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://ESO/Plate#ESO.R.444</TD>
      <TD>ivo://ESO</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://ESO/Plate#ESO.R.445</TD>
      <TD>ivo://ESO</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://ESO/Plate#ESO.R.446</TD>
      <TD>ivo://ESO</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://ESO/Plate#ESO.R.447</TD>
      <TD>ivo://ESO</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/ESO-R</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/ESO-R</TD>
      <TD>buga</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/SERC-J</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/SERC-J</TD>
      <TD>buga</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/ESO-MAMA</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/ESO-MAMA</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/ESO-MAMA</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/SERC-MAMA</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/SERC-MAMA</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/SERC-MAMA</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
    </TABLEDATA>
   </DATA>
  </TABLE> 
  <!-- end of table DM 2018-Jan- 29 -->
  
  <TABLE name="WasGeneratedByDescription">
  <!-- relation between an Entity and the Activity that produced it  -->
   <FIELD ID="wgbd_id" arraysize="*" datatype="char" name="wgbd_id" ucd="meta.id" utype="voprov:WasGeneratedByDescription.id"/>
   <FIELD ID="wgbd_idcreatedentity" arraysize="*" datatype="char" name="wgbd_entity_id" ucd="meta.id" utype="voprov:WasGeneratedByDescription.entity"/>
   <FIELD ID="wgbd_idcreatoractivity" arraysize="*" datatype="char" name="wgbd_activity_id" ucd="meta.id" utype="voprov:WasGeneratedByDescription.activity"/>
   <FIELD ID="wgbd_role" arraysize="*" datatype="char" name="wgb_role" ucd="meta.code.class" utype="voprov:WasGeneratedByDescription.Role"/>
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>color#AlaRGB</TD>
      <TD>color#RGB</TD>
      <TD>AlaRGB</TD>
      <TD>voprov:output</TD>
    </TR>
     <TR>
      <TD>DSS2_MAMA#cds_cutout</TD>
      <TD>cutout#DSS2_MAMA</TD>
      <TD>cds_cutout</TD>
      <TD>voprov:output</TD>
    </TR>
     <TR>
      <TD>DSS2#stsciNum</TD>
      <TD>Num#DSS2</TD>
      <TD>stsciNum</TD>
      <TD>voprov:output</TD>
    </TR>
     <TR>
      <TD>MAMA#MAMANUM</TD>
      <TD>Num#MAMA</TD>
      <TD>MAMANUM-xyz</TD>
      <TD>voprov:output</TD>
    </TR>
     <TR>
      <TD>HiPS#genM</TD>
      <TD>Num#MAMA</TD>
      <TD>HipsgenM</TD>
      <TD>voprov:output</TD>
    </TR>
    </TABLEDATA>
   </DATA>
  </TABLE> 
  <TABLE name="WasGeneratedBy">
  <!-- relation between an Entity and the Activity that produced it  -->
  <FIELD ID="wgb_idcreatedentity" arraysize="*" datatype="char" name="wgb_entity_id" ucd="meta.id" utype="voprov:WasGeneratedBy.entity"/>
   <FIELD ID="wgb_idcreatoractivity" arraysize="*" datatype="char" name="wgb_activity_id" ucd="meta.id" utype="voprov:WasGeneratedBy.activity"/>
 <FIELD ID="wgb_wgbDescription_id" arraysize="*" datatype="char" name="wgb_wasGeneratedByDescription_id" ucd="meta.id" utype="voprov:WasGeneratedBy.Description"/>
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_NGC6946</TD>
      <TD>AlaRGB1</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M101</TD>
      <TD>AlaRGB2</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M33</TD>
      <TD>AlaRGB3</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M51</TD>
      <TD>AlaRGB4</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M81</TD>
      <TD>AlaRGB5</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M83</TD>
      <TD>AlaRGB6</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2color#RGB_M87</TD>
      <TD>AlaRGB7</TD>
      <TD>color#AlaRGB</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.644</TD>
      <TD>cds_cutoutj644</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.644</TD>
      <TD>cds_cutoutf644</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.644</TD>
      <TD>cds_cutoutn644</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.061</TD>
      <TD>cds_cutoutj061</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.061</TD>
      <TD>cds_cutoutf061</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.061</TD>
      <TD>cds_cutoutn061</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.143</TD>
      <TD>cds_cutoutj143</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.143</TD>
      <TD>cds_cutoutf143</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.143</TD>
      <TD>cds_cutoutn143</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.174</TD>
      <TD>cds_cutoutj174</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.174</TD>
      <TD>cds_cutoutf174</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.175</TD>
      <TD>cds_cutoutn175</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.270</TD>
      <TD>cds_cutoutj270</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.270</TD>
      <TD>cds_cutoutf270</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.270</TD>
      <TD>cds_cutoutn270</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.413</TD>
      <TD>cds_cutoutj413</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.413</TD>
      <TD>cds_cutoutf413</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.413</TD>
      <TD>cds_cutoutn413</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/DSS2/SERC#SERC.I-DSS2.445</TD>
      <TD>cds_cutouti445</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/SERC#SERC.J-MAMA.444</TD>
      <TD>cds_cutoutj444</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/ESO#ESO.R-MAMA.444</TD>
      <TD>cds_cutoutr444</TD>
      <TD>DSS2_MAMA#cds_cutout</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.061</TD>
      <TD>stsciNum-21</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.061</TD>
      <TD>stsciNum-22</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.061</TD>
      <TD>stsciNum-23</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.143</TD>
      <TD>stsciNum-24</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.143</TD>
      <TD>stsciNum-25</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.143</TD>
      <TD>stsciNum-26</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.270</TD>
      <TD>stsciNum-27</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.270</TD>
      <TD>stsciNum-28</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.270</TD>
      <TD>stsciNum-29</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.174</TD>
      <TD>stsciNum-2a</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.174</TD>
      <TD>stsciNum-2b</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.175</TD>
      <TD>stsciNum-2c</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.413</TD>
      <TD>stsciNum-2d</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.413</TD>
      <TD>stsciNum-2e</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.413</TD>
      <TD>stsciNum-2f</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.644</TD>
      <TD>stsciNum-2g</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.644</TD>
      <TD>stsciNum-2h</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.644</TD>
      <TD>stsciNum-2i</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://STSci/Num#SERC.I-DSS2.445</TD>
      <TD>stsciNum-2j</TD>
      <TD>DSS2#stsciNum</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#SERC.J-MAMA.444</TD>
      <TD>MAMANUM-1</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.444</TD>
      <TD>MAMANUM-2</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.445</TD>
      <TD>MAMANUM-3</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.446</TD>
      <TD>MAMANUM-4</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.447</TD>
      <TD>MAMANUM-5</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/ESO-MAMA</TD>
      <TD>MAMANUM-ESOR</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://obspm/gepi/SERC-MAMA</TD>
      <TD>MAMANUM-SERCJ</TD>
      <TD>MAMA#MAMANUM</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/ESO-R</TD>
      <TD>EHG1</TD>
      <TD>HiPS#genM</TD>
    </TR>
     <TR>
      <TD>ivo://cds/P/MAMA/SERC-J</TD>
      <TD>EHG2</TD>
      <TD>HiPS#genM</TD>
    </TR>
    </TABLEDATA>
   </DATA>
  </TABLE>
 <!-- table wgb compatible DM 2018-Jan- 29 -->
   <!-- compatible DM 2018-Jan- 29 -->
  <TABLE name="WasAssociatedWith">
   <FIELD ID="waw_activityID" arraysize="*" datatype="char" name="waw_activity_id" ucd="meta.id" utype="voprov:WasAssociatedWith.activity"/>
   <FIELD ID="waw_agentID" arraysize="*" datatype="unicodeChar" name="waw_agent_id" ucd="meta.id" utype="voprov:WasAssociatedWith.agent"/>
   <FIELD ID="waw_role" arraysize="*" datatype="char" name="waw_role" ucd="meta.code.class" utype="voprov:WasAssociatedWith.agentRole"/>
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>AlaRGB1</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB1</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>AlaRGB2</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB2</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>AlaRGB3</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB3</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>AlaRGB4</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB4</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>AlaRGB5</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB5</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>AlaRGB6</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB6</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>AlaRGB7</TD>
      <TD>bonnarel</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>AlaRGB7</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-21</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-22</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-23</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-24</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-25</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-26</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-27</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-28</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-29</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2a</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2b</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2c</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2d</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2e</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2f</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2g</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2h</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2i</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>stsciNum-2j</TD>
      <TD>ivo://stsci</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-1</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-2</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-3</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-4</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-5</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-ESOR</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-SERCJ</TD>
      <TD>ivo://gepi</TD>
      <TD>voprov:curator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-1</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-2</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-3</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-4</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-5</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-ESOR</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-SERCJ</TD>
      <TD>ivo://obspm/cai</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>MAMANUM-1</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>MAMANUM-2</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>MAMANUM-3</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>MAMANUM-4</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>MAMANUM-5</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>MAMANUM-ESOR</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>MAMANUM-SERCJ</TD>
      <TD>Guibert</TD>
      <TD>voprov:author</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj061</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf061</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn061</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj143</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf143</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn143</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj174</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf174</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn175</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj270</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf270</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn270</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj413</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf413</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn413</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj644</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf644</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn644</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>EHG1</TD>
      <TD>buga</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>EHG1</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
     <TR>
      <TD>EHG2</TD>
      <TD>buga</TD>
      <TD>voprov:operator</TD>
    </TR>
     <TR>
      <TD>EHG2</TD>
      <TD>ivo://cds</TD>
      <TD>voprov:creator</TD>
    </TR>
    </TABLEDATA>
   </DATA>
  </TABLE>
<!-- table used conform to DM on 2018-Jan-29 -->
  <TABLE name ="UsedDescription">
   <FIELD ID="ud_id" arraysize="*" datatype="char" name="ud_id" ucd="meta.id" utype="voprov:UsedDescription.id"/>
   <FIELD ID="ud_activityDescription_id" arraysize="*" datatype="char" name="ud_activityDescription_id" ucd="meta.id" utype="voprov:UsedDescription.activityDescription"/>
   <!-- foreign key to the activity id into Activity table -->
   <FIELD ID="ud_entityDescription_id" arraysize="*" datatype="char" name="ud_entityDescription_id" ucd="meta.id" utype="voprov:UsedDescription.entityDescription"/>
   <!-- foreign key to the Entity table -->
   <FIELD ID="ud_role" arraysize="*" datatype="char" name="ud_role" ucd="meta.code.class" utype="voprov:UsedDescription.Role"/>
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>AlaRGB#cutout</TD>
      <TD>AlaRGB</TD>
      <TD>cutout#DSS2_MAMA</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>cutout#NumDSS2</TD>
      <TD>cds_cutout</TD>
      <TD>Num#DSS2</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>cutout#NumMAMA</TD>
      <TD>cds_cutout</TD>
      <TD>Num#MAMA</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>stsciNum#PlPOSSII</TD>
      <TD>stciNum</TD>
      <TD>Plate#POSSII</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>stsciNum#PlSERC</TD>
      <TD>stsciNum</TD>
      <TD>Plate#SERC</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>MAMANUM#PlSERC</TD>
      <TD>MAMANUM-xyz</TD>
      <TD>Plate#SERC</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>MAMANUM#PlESO</TD>
      <TD>MAMANUM-xyz</TD>
      <TD>Plate#ESO</TD>
      <TD>voprov:input</TD>
    </TR>
     <TR>
      <TD>HiPSigenM#numMAMA</TD>
      <TD>HipsgenM</TD>
      <TD>num#MAMA</TD>
      <TD>voprov:input</TD>
    </TR>
    </TABLEDATA>
   </DATA>
</TABLE>
<TABLE name ="Used">
   <!-- foreign key to the activity id into Activity table -->
   <FIELD ID="use_idusedactivity" arraysize="*" datatype="char" name="u_activity_id" ucd="meta.id" utype="voprov:Used.activity"/>
   <FIELD ID="use_iduseentity" arraysize="*" datatype="char" name="u_entity_id" ucd="meta.id" utype="voprov:Used.entity"/>
   <!-- foreign key to the Entity table -->
   <FIELD ID="u_usedDescription_id" arraysize="*" datatype="char" name="u_usedDescription_id" ucd="meta.id" utype="voprov:Used.usedDescription"/>
   <DATA>
    <TABLEDATA>
     <TR>
      <TD>AlaRGB1</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.143</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB1</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.143</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB1</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.143</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB2</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.174</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB2</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.174</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB2</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.175</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB3</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.413</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB3</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.413</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB3</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.413</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB4</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.270</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB4</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.270</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB4</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.270</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB5</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.061</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB5</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.061</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB5</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.061</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB6</TD>
      <TD>ivo://cds/P/MAMA/SERC#SERC.J-MAMA.444</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB6</TD>
      <TD>ivo://cds/P/MAMA/ESO#ESO.R-MAMA.444</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB6</TD>
      <TD>ivo://cds/P/DSS2/SERC#SERC.I-DSS2.445</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB7</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.644</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB7</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.644</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>AlaRGB7</TD>
      <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.644</TD>
      <TD>AlaRGB#cutout</TD>
    </TR>
     <TR>
      <TD>MAMANUM-1</TD>
      <TD>ivo://SERC/Plate#SERC.J.444</TD>
      <TD>MAMANUM#PlSERC</TD>
    </TR>
     <TR>
      <TD>MAMANUM-2</TD>
      <TD>ivo://ESO/Plate#ESO.R.444</TD>
      <TD>MAMANUM#PlESO</TD>
    </TR>
     <TR>
      <TD>MAMANUM-3</TD>
      <TD>ivo://ESO/Plate#ESO.R.445</TD>
      <TD>MAMANUM#PlESO</TD>
    </TR>
     <TR>
      <TD>MAMANUM-4</TD>
      <TD>ivo://ESO/Plate#ESO.R.446</TD>
      <TD>MAMANUM#PlESO</TD>
    </TR>
     <TR>
      <TD>MAMANUM-5</TD>
      <TD>ivo://ESO/Plate#ESO.R.447</TD>
      <TD>MAMANUM#PlESO</TD>
    </TR>
     <TR>
      <TD>stsciNum-21</TD>
      <TD>ivo://POSSII/Plate#POSSII.J.061</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-22</TD>
      <TD>ivo://POSSII/Plate#POSSII.F.061</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-23</TD>
      <TD>ivo://POSSII/Plate#POSSII.N.061</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-24</TD>
      <TD>ivo://POSSII/Plate#POSSII.J.143</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-25</TD>
      <TD>ivo://POSSII/Plate#POSSII.F.143</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-26</TD>
      <TD>ivo://POSSII/Plate#POSSII.N.143</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2b</TD>
      <TD>ivo://POSSII/Plate#POSSII.F.174</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2c</TD>
      <TD>ivo://POSSII/Plate#POSSII.N.175</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2a</TD>
      <TD>ivo://POSSII/Plate#POSSII.J.174</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-27</TD>
      <TD>ivo://POSSII/Plate#POSSII.J.270</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-29</TD>
      <TD>ivo://POSSII/Plate#POSSII.N.270</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-28</TD>
      <TD>ivo://POSSII/Plate#POSSII.F.270</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2d</TD>
      <TD>ivo://POSSII/Plate#POSSII.J.413</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2e</TD>
      <TD>ivo://POSSII/Plate#POSSII.F.413</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2f</TD>
      <TD>ivo://POSSII/Plate#POSSII.N.413</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2g</TD>
      <TD>ivo://POSSII/Plate#POSSII.J.644</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2h</TD>
      <TD>ivo://POSSII/Plate#POSSII.F.644</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2i</TD>
      <TD>ivo://POSSII/Plate#POSSII.N.644</TD>
      <TD>stsciNum#PlPOSSII</TD>
    </TR>
     <TR>
      <TD>stsciNum-2j</TD>
      <TD>ivo://SERC/Plate#SERC.I.445</TD>
      <TD>stsciNum#PlSERC</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj061</TD>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.061</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf061</TD>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.061</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn061</TD>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.061</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj644</TD>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.644</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf644</TD>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.644</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn644</TD>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.644</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj143</TD>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.143</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf143</TD>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.143</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn143</TD>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.143</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj174</TD>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.174</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf174</TD>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.174</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn175</TD>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.175</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj270</TD>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.270</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf270</TD>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.270</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn270</TD>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.270</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj413</TD>
      <TD>ivo://STSci/Num#POSSII.J-DSS2.413</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutf413</TD>
      <TD>ivo://STSci/Num#POSSII.F-DSS2.413</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutn413</TD>
      <TD>ivo://STSci/Num#POSSII.N-DSS2.413</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>cds_cutoutj444</TD>
      <TD>ivo://gepi/MAMA/Num#SERC.J-MAMA.444</TD>
      <TD>cutout#NumMAMA</TD>
    </TR>
     <TR>
      <TD>cds_cutoutr444</TD>
      <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.444</TD>
      <TD>cutout#NumMAMA</TD>
    </TR>
     <TR>
      <TD>cds_cutouti445</TD>
      <TD>ivo://STSci/Num#SERC.I-DSS2.445</TD>
      <TD>cutout#NumDSS2</TD>
    </TR>
     <TR>
      <TD>EHG1</TD>
      <TD>ivo://obspm/gepi/ESO-MAMA</TD>
      <TD>HiPSigenM#numMAMA</TD>
    </TR>
     <TR>
      <TD>EHG2</TD>
      <TD>ivo://obspm/gepi/SERC-MAMA</TD>
      <TD>HiPSigenM#numMAMA</TD>
   </TR>
  </TABLEDATA>
 </DATA>
</TABLE>
<TABLE name="params" utype="voprov:parameter" >
	<FIELD name ="p_isaParamOf" utype="voprov:Parameter.activity" ucd="meta.id" datatype="char" arraysize="*" />
	<FIELD name="p_id" utype="voprov:Parameter.id"  datatype="char" ucd="meta.id" arraysize="*" />
	<FIELD name="p_value" utype="voprovPparameter.value"  ucd="stat;value" datatype="char" arraysize="*" />
	<FIELD name="p_parameterDescription" utype="voprov:parameter.parameterDescription" ucd="meta.id" datatype="char" arraysize="*" />
  <DATA>
    <TABLEDATA>
	<TR>
            <TD>EHG1</TD>
            <TD>hips_tile_format1</TD>
            <TD>jpeg fits</TD>
            <TD>hips_tile_format</TD>
         </TR>
         <TR>
            <TD>EHG1</TD>
            <TD>hips_order1</TD>
            <TD>6</TD>
            <TD>hips_order</TD>
         </TR>
         <TR>
            <TD>EHG2</TD>
            <TD>hips_tile_format2</TD>
            <TD>fits</TD>
            <TD>hips_tile_format</TD>
         </TR>
         <TR>
            <TD>EHG2</TD>
            <TD>hips_order2</TD>
            <TD>5</TD>
            <TD>hips_order</TD>
         </TR>
    </TABLEDATA>
  </DATA>
</TABLE>
<TABLE name="parameterDescription" utype="voprov:parameterDescription" >
<FIELD name ="isaParamOf" utype="voprov:parameterDescription.activityDescription" ucd="meta.id" datatype="char" arraysize="*" />
<FIELD name="id" utype="voprov:parameterDescription.id"  ucd="meta.id" datatype="char" arraysize="*" />
<FIELD name="unit" utype="voprov:parameterDescription.unit"  ucd="meta.unit" datatype="char" arraysize="*" />
<FIELD name="ucd" utype="voprov:parameterDescription.ucd"  ucd="meta.ucd" datatype="char" arraysize="*" />
  <DATA>
    <TABLEDATA>
         <TR>
            <TD>HipsGenM</TD>
            <TD>hips_tile_format</TD>
            <TD></TD>
            <TD>meta.code.class</TD>
         </TR>
         <TR>
            <TD>HipsGenM</TD>
            <TD>hips_order</TD>
            <TD></TD>
            <TD>meta.code</TD>
         </TR>
    </TABLEDATA>
  </DATA>
</TABLE>
<TABLE name="miniobscore" utype="obscore:" >
       <FIELD name="obs_publisher_did" ucd="meta.ref.ivoid" utype="obscore:Curation.publisherDID" datatype = "char" arraysize="*" />
       <FIELD name="access_url" ucd="meta.ref.url" utype="obscore:Access.reference" datatype="char" arraysize="*" />
	   <FIELD name="data_rights" ucd="meta.code.class" utype="obscore:Curation.rights" datatype="char" arraysize="*" />
       <FIELD name="dataproduct_type" ucd="meta.code.class" utype="obscore:ObsDataSet.dataProductype"
 datatype="char" arraysize="*"  />
       <FIELD name="calib_level" ucd="meta.code;obs.calib" utype="obscore:ObsDataset.calibLevel" datatype="int"   />
  <DATA>
     <TABLEDATA>
#added public in data rights 
--! 7 color images -->
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_NGC6946</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_NGC6946.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M101</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_M101.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M33</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_M33.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M51</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_M51.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M81</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_M81.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M83</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_M83.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2color#RGB_M87</TD>
            <TD>file://home/bonnarel/Aladin/RGB/RGB_M87.png</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>3</TD>
         </TR>
--! 21 cutouts -->
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.061</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.N-DSS2.061.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.061</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.J-DSS2.061.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.061</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.F-DSS2.061.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.143</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.J-DSS2.143.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.143</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.F-DSS2.143.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.143</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.N-DSS2.143.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.174</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.J-DSS2.174.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.174</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.F-DSS2.174.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.175</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.N-DSS2.175.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.270</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.J-DSS2.270.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.270</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.F-DSS2.270.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.270</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.N-DSS2.270.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.413</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.J-DSS2.413.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.413</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.F-DSS2.413.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.413</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.N-DSS2.413.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.J-DSS2.644</TD>
            <TD>POSSII Blue Survey DSS2 M87</TD>
            <TD></TD>
            <TD>This is the cut-out DSS2 digitization of the Blue POSSII Schmidt survey around Messier 87</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.J-DSS2.644.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.F-DSS2.644</TD>
            <TD>POSSII Red Survey DSS2 M87</TD>
            <TD></TD>
            <TD>This is the cut-out DSS2 digitization of the Red POSSII Schmidt survey around Messier 87</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.F-DSS2.644.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/POSSII#POSSII.N-DSS2.644</TD>
            <TD>POSSII Infra Red Survey DSS2 M87</TD>
            <TD></TD>
            <TD>This is the cut-out DSS2 digitization of the Infra red POSSII Schmidt survey around Messier 87</TD>
            <TD>file://home/bonnarel/Aladin/RGB/POSSII.N-DSS2.644.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/DSS2/SERC#SERC.I-DSS2.445</TD>
            <TD>file://home/bonnarel/Aladin/RGB/SERC.I-DSS2.445.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/MAMA/SERC#SERC.J-MAMA.444</TD>
            <TD>file://home/bonnarel/Aladin/RGB/SERC.J-MAMA.444.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://cds/P/MAMA/ESO#ESO.R-MAMA.444</TD>
            <TD>file://home/bonnarel/Aladin/RGB/ESO.R-MAMA.444.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>2</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.061</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.J-DSS2.061.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.061</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.F-DSS2.061.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.061</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.N-DSS2.061.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.143</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.J-DSS2.143.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.143</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.F-DSS2.143.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.143</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.N-DSS2.143.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.174</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.J-DSS2.174.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.174</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.F-DSS2.174.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.175</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.N-DSS2.175.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.270</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.J-DSS2.270.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.270</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.F-DSS2.270.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.270</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.N-DSS2.270.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.413</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.J-DSS2.413.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.413</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.F-DSS2.413.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.413</TD>
            <TD>http://stsci.edu/dss2/possii/POSSII.N-DSS2.413.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.J-DSS2.644</TD>
            <TD>NULL</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.F-DSS2.644</TD>
            <TD>NULL</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#POSSII.N-DSS2.644</TD>
            <TD>NULL</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://STSci/Num#SERC.I-DSS2.445</TD>
            <TD>NULL</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://gepi/MAMA/Num#SERC.J-MAMA.444</TD>
            <TD>http://stsci.edu/dss2/possii/SERC.J-MAMA.444.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
         </TR>
         <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.444</TD>
            <TD>http://stsci.edu/dss2/possii/ESO.R-MAMA.444.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
            <TD>Coll-ESO-R-Key</TD>
         </TR>
         <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.445</TD>
            <TD>http://stsci.edu/dss2/possii/ESO.R-MAMA.445.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
            <TD>Coll-ESO-R-Key</TD>
         </TR>
         <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.446</TD>
            <TD>http://stsci.edu/dss2/possii/ESO.R-MAMA.446.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
            <TD>Coll-ESO-R-Key</TD>
         </TR>
         <TR>
            <TD>ivo://gepi/MAMA/Num#ESO.R-MAMA.447</TD>
            <TD>http://stsci.edu/dss2/possii/ESO.R-MAMA.447.fits</TD>
            <TD>public</TD>
            <TD>image</TD>
            <TD>1</TD>
            <TD>Coll-ESO-R-Key</TD>
         </TR>
--! Shmidt photo plates -->
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.644</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.644</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.644</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.061</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.061</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.061</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.143</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.143</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.143</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.174</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.174</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.175</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.270</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.270</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.270</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.J.413</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.F.413</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://POSSII/Plate#POSSII.N.413</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://SERC/Plate#SERC.I.445</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://SERC/Plate#SERC.J.444</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://ESO/Plate#ESO.R.444</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://ESO/Plate#ESO.R.445</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://ESO/Plate#ESO.R.446</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
         <TR>
            <TD>ivo://ESO/Plate#ESO.R.447</TD>
            <TD>NULL</TD>
            <TD>reserved</TD>
            <TD>plate</TD>
            <TD>0</TD>
         </TR>
    </TABLEDATA>
  </DATA>
</TABLE>
</RESOURCE>
</VOTABLE>
<!-- entity description table -->
\end{lstlisting}


\section{Changes from Previous Versions}
mireille: changed stopTime everywhere in endTime and ucd ``time.stop'' in ``time.end''  to ensure compatibility with provDM tables. 
\end{appendices}
\bibliography{ivoatex/ivoabib,prov-refs}


\end{document}
